Author’s response to reviews

Title: A Potential Risk Factor of Total Knee Arthroplasty: An Infected Baker’s Cyst – A Case Report

Authors:

Byung-il Lee (bilee@schmc.ac.kr)
Jong-Hyeon Seo (111159@schmc.ac.kr)
Yong-Beom Kim (schkyb@schmc.ac.kr)
Gi-won Seo (102980@schmc.ac.kr)

Version: 3 Date: 18 Feb 2020

Author’s response to reviews:

Editor Comments:

1. The title in this manuscript is duplicated. Please amend this.

We amended the duplicated title.

2. Please amend your statement in the "Consent for publication" section, to clarify whether written informed consent for publication was provided by the patient.

We changed the sentence like below.

The patient agreed to the publication of this report and we have attached a consent file
☞ The patient provided informed consent for publication of this case report and accompanying images.
3. Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

We changed the manuscript like below.

Funding
Not applicable
☞ This work was supported by the Soonchunhyang University Research Fund.

Acknowledgement
This work was supported by the Soonchunhyang University Research Fund.
☞ Not applicable

4. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of author BIL do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their name from the list of authors and place them in the “Acknowledgements” section instead.

Author’s contributions has been modified

All authors have read and approved the manuscript.

(1) The concept and design of the study: GWS
(2) Drafting the article or revising it critically for important intellectual content: JHS, YBK
(3) Final approval of the version to be submitted: BIL
All authors have read and approved the manuscript.

1. The concept and design of the study, data collection and analysis: BIL, GWS
2. Drafting the article or revising it critically for important intellectual content: JHS, YBK
3. Final approval of the version to be submitted: BIL

5. Please remove the response to reviewers' comments letter, as it is no longer needed at this stage. Please do not upload a new letter with this revision.

We removed the review letter from Supplementary material.

6. Now that peer-review is complete, please remove the CARE checklist from the file inventory. Please add a statement to the manuscript that you adhered to CARE guidelines/methodology.

We have removed the CARE checklist.

7. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

We edited as you recommended.